Author's response to reviews

Title: Segregation analysis in families with Charcot-Marie-Tooth disease allows reclassification of putative disease causing mutations.

Authors:

Rune Ostern (runeostern@online.no)
Toril Fagerheim (toril.fagerheim@unn.no)
Helene Hjellnes (helene.hjellnes@unn.no)
Bjorn Nygard (bjorn.nygard@unn.no)
Svein I Mellgren (svein.ivar.mellgren@uit.no)
Oivind Nilssen (oivind.nilssen@uit.no)

Version: 2 Date: 4 November 2013

Author's response to reviews:

Dear Jesus Ervin Cenzon
Journal Editorial Office
BioMed Central

Regarding MS: 9711485801120118

Segregation analysis in families with Charcot-Marie-Tooth disease allows reclassification of putative disease causing mutations.
Rune Ostern, Toril Fagerheim, Helene Hjellnes, Bjorn Nygard, Svein I Mellgren and Oivind Nilssen

Here is a point-by-point list of changes made to the manuscript:

1: The email address of all authors is included in the title page.

2: Data collection, statistics and endpoint measures, Page 7, line 14-17:
The sentences “The study was approved by the Regional Committee for Medical Research Ethics and the Norwegian Data Inspectorate. Procedures were in accordance with the revised Helsinki Declaration of 2008.” have been changed to “The study was approved by the Norwegian Data Inspectorate and the Regional Committee for Medical Research Ethics who specified that it was not necessary with informed consent given by subjects/next-of-kin for this study. Procedures were in accordance with the revised Helsinki Declaration of 2008.”

With best regards
Rune Østern, corresponding author